

A Spherical Harmonic Model of Earth's Lithospheric Magnetic Field up to Degree 1050

Erwan Thébault, Gauthier Hulot, Benoit Langlais, Pierre Vigneron

► To cite this version:

Erwan Thébault, Gauthier Hulot, Benoit Langlais, Pierre Vigneron. A Spherical Harmonic Model of Earth's Lithospheric Magnetic Field up to Degree 1050. Geophysical Research Letters, 2021, 48 (21), 10.1029/2021GL095147 . hal-03420984

HAL Id: hal-03420984 https://nantes-universite.hal.science/hal-03420984

Submitted on 15 Nov 2021

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution - NonCommercial 4.0 International License

ഹ



Geophysical Research Letters^{*}

RESEARCH LETTER

10.1029/2021GL095147

Key Points:

- Model based on a joint inversion of Swarm and CHAMP satellites magnetic field measurements and near-surface scalar anomaly gridded data
- First global lithospheric vector magnetic field model at the Earth's mean radius to about 40 km resolution

Correspondence to: E. Thébault, erwan.thebault@uca.fr

erwan.mebaun@uea.

Citation:

Thébault, E., Hulot, G., Langlais, B., & Vigneron, P. (2021). A spherical harmonic model of Earth's lithospheric magnetic field up to degree 1050. *Geophysical Research Letters*, 48, e2021GL095147. https://doi. org/10.1029/2021GL095147

Received 12 JUL 2021 Accepted 13 OCT 2021

© 2021. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

A Spherical Harmonic Model of Earth's Lithospheric Magnetic Field up to Degree 1050

E. Thébault¹, G. Hulot², B. Langlais³, and P. Vigneron²

¹Laboratoire Magma et Volcans, Université Clermont Auvergne, CNRS, IRD, OPGC, Clermont-Ferrand, France, ²Université de Paris, Institut de Physique du Globe de Paris, CNRS, Paris, France, ³Laboratoire de Planétologie et Géodynamique, UMR 6112, Université de Nantes, Université d'Angers, Nantes, France

Abstract Detailed mapping of the Earth's magnetic field brings key constraints on the composition, dynamics, and history of the crust. Satellite and near-surface measurements detect different length scales and are complementary. Here, we build a model, selecting and processing magnetic field measurements from the German CHAMP and ESA Swarm satellites, which we merge with near-surface scalar anomaly data. We follow a regional approach for modeling the magnetic field measurements and next transform the series of regional models into a unique set of spherical harmonic (SH) Gauss coefficients. This produces the first global model to SH degree 1050 derived by inversion of all available measurements. The new model agrees with previous satellite-based models at large wavelengths and fits the CHAMP and Swarm satellite data down to expected noise levels. Further assessment in the geographical and spectral domains shows the model to be stable when downward continued to the Earth's surface.

Plain Language Summary The magnetic field of the Earth's results from the superposition of various internal and external fields. The field produced by magnetized rocks in the Earth's lithosphere brings essential constraints on the crust composition, dynamics, and history. Its description requires magnetic measurements in all regions of the world at different altitudes. Satellite measurements detect well the large Earth's magnetic field structures but are too far to detect the smaller ones that are weak at high altitudes. Historical marine and airplane data can be used to complement the satellite observations. Considering all these data makes it possible to build a global model that represents the magnetic field generated within the Earth's crust with a global 40 km horizontal spatial resolution.

1. Introduction

The magnetic field produced by the Earth's lithosphere arises because of lateral contrasts between magnetized rocks (e.g., Purucker & Clark, 2011). It carries information about the magnetization and composition of these rocks, the age and contours of geological blocks at depth, and more generally the structure, thermal regime, and tectonic evolution of the Earth's crust (e.g., Purucker et al., 2002). Deciphering the geographical variations of the lithospheric magnetic field is also important for navigation purposes and requires consistent data over distances spanning continental boundaries and covering oceans from the near-surface to satellite altitudes.

Globally consistent models derived from measurements at satellite altitudes, using the German CHAMP (Reigber et al., 2002) and ESA Swarm (Friis-Christensen et al., 2006) satellites, have a spatial horizontal resolution limited to 250 km (Olsen et al., 2017). Complementary data acquired at much lower altitude are publicly available as regional marine and aeromagnetic surveys and compilations. These data are more sensitive to small-scale structures and allow spatial resolutions ranging from 5 to 50 km to be reached (see Maus et al., 2007, their Table 1). However, because of their limited spatial extent, these surveys poorly document large-scale field structures (e.g., Ravat et al., 2002). Reconciling the vector satellite and the near-surface scalar measurements is required to depict the Earth's lithospheric magnetic field in a globally consistent way.

The World Digital Magnetic Anomaly Map (WDMAM) project was a first international effort toward that goal. First released in 2007 (Korhonen et al., 2007), the WDMAM was later updated (Lesur et al., 2016) using more extensive near-surface data coverage, improved data processing in the oceanic domain (Dyment et al., 2015), and more consistent merging of the individual regional datasets. An independent satellite-based model was also used to incorporate information about large spatial scales, by downward continuation to

4 km altitude above the Earth's ellipsoid. Alternative global scalar anomaly grids, such as the Earth Magnetic Anomaly Grid at 2 arc minutes resolution (EMAG2, Maus et al., 2009) and its latest update, the EMAG2v3 grid, were also constructed along the same lines (Meyer et al., 2017).

Both WDMAM and EMAG2 scalar anomaly grids were next converted into sets of spherical harmonic (SH) Gauss coefficients, using linearization and regularization (Lesur et al., 2016; Maus, 2010) to recover models of the vector magnetic field. However, no global inversion combining all available near-surface gridded data and satellite direct measurements have yet been performed to best take advantage of the data and build a fully consistent joint model.

Here, we carry such a joint inversion, using CHAMP and Swarm measurements combined with the WD-MAM-2 anomaly grid to build a model of the Earth's lithospheric magnetic vector field with a 40-km horizontal spatial resolution. For the first time, all measurements and data types (satellite vector, scalar, and gradient data, as well as aeromagnetic and marine compilations) are simultaneously used. The method involves an iterative piecewise approach using the Revised Spherical Cap Harmonic functions (R-SCHA, Thébault, 2006) to allow independent regional data analyses, build regional models and combine them into a global SH model. This makes it possible to best take advantage of all available measurements at reduced and manageable numerical costs.

This study is organized as follows. Section 2 describes the CHAMP and Swarm satellite data, the selection and processing methods we use, and provides a summary of the WDMAM scalar grid. Section 3 describes the regional mathematical representation of the magnetic field and the inversion approach. Inversion statistics are next presented in Section 4, where we also assess and discuss the final SH model, which we compare to earlier models.

2. Data Selection and Correction

2.1. Swarm and CHAMP Satellite Data Selection

We use 1 Hz Level 1b vector and scalar measurements of the lowest Swarm A and C satellites between March 1, 2014 and March 31, 2020, and between 430 and 460 km altitude. Data are selected only when the instruments are in nominal mode and when the thrusters are off. They are next sub-sampled every 5 s, which corresponds to about 35 km along-track spacing at satellite altitude.

Only magnetic field measurements corresponding to magnetospheric quiet times are used. These are identified by the planetary Kp index being lower than 2° in value both at time of measurement and during the previous three hours. This selection is refined by further requesting the Disturbed Storm Time Index (Dst) to not exceed 5 nT in absolute value and not vary by more than 5 nT during the three previous hours. We additionally require the interplanetary magnetic field (IMF) to be less than 5 nT in absolute value for the *y*-component and positive for the *z*-component, for a better rejection of measurements corresponding to precursors of strong events such as geomagnetic storms and substorms (Richardson & Cane, 2011).

The statistical properties of the remaining perturbation fields after selection are different at low and high latitudes. We separate the selected data set into midlatitudes (between -60° and 60° magnetic latitudes MLAT, as defined by the Quasi-Dipole coordinates at satellite altitude, Richmond, 1995) and polar-latitude regions (between 50° and 90° and -90° and -50° MLAT for the north and south caps, respectively). The polar and midlatitude subsets overlap by 10° in latitude. This precaution allows more robust further dedicated data corrections, mitigates edge effects, and ensures a better continuity between mid and polar regions (see also Thébault et al., 2013 for the algorithm description). Nighttime measurements between 21:00 and 5:00 local time (LT) are kept at midlatitudes to minimize contamination from the pseudo-periodic diurnal ionospheric Sq field (see Chulliat et al., 2013, for instance). In the polar regions, the measurements are taken in the dark and when the Sun is at least 10° below the horizon.

We also include a selection of the latest calibrated CHAMP measurements (GFZ Section 2.3, 2019) from January 2001 to September 2010 between 250 and 480 km altitude. As is done for the Swarm data, the CHAMP data are split into mid and high magnetic latitude datasets with a 10° overlap between regions. At midlatitudes, the measurements are selected between 22:00 and5:00 LT and all local times are considered in the polar areas with the condition that the satellite was in the dark with the Sun located at least 5° below

the horizon. A final data selection is made against the Kp and the Dst magnetic indices and the IMF values following the same criteria as those considered for the Swarm data.

2.2. Satellite Data Correction

We correct the Swarm and the CHAMP measurements for the internal core field and its temporal variations up to SH degree 15 and for the external primary and internal induced magnetospheric fields using the CHAOS-7.5 model (Finlay et al., 2020). This model covers the full CHAMP and Swarm data epochs and was derived with the same calibration versions of the CHAMP and Swarm data. After these corrections, vector and scalar gradient data sets are also built. For Swarm, across track gradients are constructed from satellites A and C by selecting pairs of data measured at the same Universal Times at all latitudes. For CHAMP and Swarm, along-track gradients are constructed considering 15 s time spacing between measurements along the satellite orbits, following Finlay et al. (2020).

The corrected data show large-scale remaining structures at mid-latitudes and significant transient perturbations in the polar areas blurring the lithospheric field signal. At mid-latitudes, these signals are caused by rapid magnetospheric and ionospheric fields not accounted for by the CHAOS-7.5 model correction. In polar areas, perturbations along satellite orbits are mostly due to the magnetic field of Polar Electrojet and Field Aligned Currents (FAC). This calls for dedicated corrections using additional filters along the Swarm and CHAMP satellite trajectories.

At midlatitudes, the perturbation field contaminates mostly the zonal terms (Maus et al., 2008). To deal with this, we next correct the data along each satellite track in the geomagnetic reference frame for an external SH degree 1 field and its internal induced part, and an additional SH degree two external zonal term. In polar regions, the transient external fields are more effectively reduced with a non-parametric filter such as the Singular Spectral Analysis (SSA; see Thébault et al., 2017). For the SSA, the perturbation field is mostly carried by the first two singular values of the SSA. A final screening is performed to manually reject satellite track measurements with obvious first-order issues such as spikes or outlier values.

After all corrections, only measurements between 55° and 90°, and between -90° and -55° MLAT for the north and south caps, respectively, and between -55° and 55° MLAT at midlatitudes are kept, thus removing the overlap areas between mid and high magnetic latitudes regions that were considered to minimize edge effects between regions. The contributions of the lithospheric field magnetic field to SH degree 80 are then added back to all data.

2.3. Near-Surface Data

Several near-surface magnetic anomaly compilations are publicly available. The EMAG2 scalar anomaly grid (Maus et al., 2009) is not considered in this work as it misses some recent regional updated compilations. Its latest version, the EMAG2v3 (Meyer et al., 2017) is not considered either, since parts of oceanic areas contain wide regional data voids and sharp transitions. To minimize numerical instabilities and spatial ringing in our model, we select the WDMAM-2 grid that offers the most extensive geographical coverage (Lesur et al., 2016). About 3.5% of the data sets are synthetic but built from reliable and carefully calibrated geophysical priors (Dyment et al., 2015). As is the case for all global scalar anomaly grids, the large wavelengths are imposed by a satellite-based model. In the WDMAM-2, the structures corresponding to SH degrees between 16 and 100 originate from the GRIMM_L120 lithospheric field model downward continued to 5 km altitude (Lesur et al., 2013). These large-scale structures might not be fully compatible Table 1

Residual Statistics Between the SH Model and the Initial Data Sets for CHAMP, Swarm, and WDMAM Data

| | Swarm | | | СНАМР | | | WDMAM | | |
|---------------------------|-----------|-------|------|-----------|-------|------|-----------|-------|-------|
| | N | Mean | rms | N | Mean | rms | N | Mean | rms |
| Fpolar (nT) | 3,181,999 | 0.02 | 2.91 | 945,958 | -0.76 | 4.01 | 588,024 | -1.77 | 22.91 |
| Br,polar (nT) | 3,181,999 | -0.03 | 2.43 | 945,958 | 0.11 | 2.58 | 0 | N/A | N/A |
| B0.polar (nT) | 3,181,999 | 0.11 | 3.58 | 945,958 | 0.24 | 3.71 | 0 | N/A | N/A |
| Bø.polar (nT) | 3,181,999 | -0.14 | 3.72 | 945,958 | 0.16 | 3.08 | 0 | N/A | N/A |
| F non-polar (nT) | 4,600,571 | 0.01 | 1.2 | 3,045,208 | -0.29 | 1.45 | 2,651,976 | -2.16 | 18.51 |
| Br. non-polar (nT) | 4,600,571 | 0.00 | 0.91 | 3,045,208 | 0.11 | 1.23 | 0 | N/A | N/A |
| Bθ. non-polar (nT) | 4,600,571 | 0.08 | 1.12 | 3,045,208 | 0.12 | 2.14 | 0 | N/A | N/A |
| Bφ. non-polar (nT) | 4,600,571 | 0.02 | 1.14 | 3,045,208 | 0.1 | 2.06 | 0 | N/A | N/A |
| δF-AT. polar (nT) | 796,122 | 0.01 | 0.47 | 236,490 | 0.01 | 0.52 | 0 | N/A | N/A |
| δF-AT. non-polar (nT) | 796,122 | -0.01 | 0.15 | 761,302 | 0.00 | 0.34 | 0 | N/A | N/A |
| δBr-AT (nT) | 796,122 | 0.00 | 0.25 | 761,302 | 0.01 | 0.32 | 0 | N/A | N/A |
| $\delta B\theta$ -AT (nT) | 796,122 | -0.01 | 0.26 | 761,302 | -0.01 | 0.29 | 0 | N/A | N/A |
| δBφ-AT (nT) | 796,122 | -0.01 | 0.23 | 761,302 | 0.00 | 0.32 | 0 | N/A | N/A |
| δF-EW. polar (nT) | 1,762,358 | 0.09 | 0.67 | 0 | N/A | N/A | 0 | N/A | N/A |
| δF-EW. non-polar (nT) | 1,762,358 | 0.03 | 0.42 | 0 | N/A | N/A | 0 | N/A | N/A |
| δBr -EW (nT) | 1,762,358 | -0.01 | 0.29 | 0 | N/A | N/A | 0 | N/A | N/A |
| $\delta B\theta$ -EW (nT) | 1,762,358 | -0.02 | 0.33 | 0 | N/A | N/A | 0 | N/A | N/A |
| $\delta B \phi$ -EW(nT) | 1,762,358 | 0.01 | 0.41 | 0 | N/A | N/A | 0 | N/A | N/A |

with our selected and processed measurements. We describe in the next section an approach designed to minimize this issue.

The WDMAM-2 grid is provided as a grid with a 5 \times 5 km cell size at 5 km altitude above the WGS84 geoid (Kovalevsky et al., 2012). We reduce the 25,927,200 total intensity anomaly data of the initial grid to about a 0.15 \times 0.15° cell size equal area grid using a cubic B-spline interpolation (about 15 km spacing). We are therefore left with 3,240,000 intensity anomaly values converted from the WGS84 geodetic to the geographic Earth-Centered Earth-Fixed (ECEF) coordinate system. This reduction procedure is necessary to avoid deriving a lithospheric magnetic field model too biased toward near-surface measurements.

3. Data Inversion and Statistics

The scalar WDMAM-2 anomaly data and the CHAMP and Swarm vector, scalar, and gradient data (see Table 1) are regionally modeled with the R-SCHA mathematical functions. The Earth is tiled with 700 caps placed on the nodes of an equal area grid. A description of the way this procedure was applied globally on the CHAMP measurements (third calibration) is provided and illustrated in Thébault (2006, their Figure 1) and the way the algorithm has been extended for taking advantage of gradient data is detailed in Thébault et al. (2013). The same general procedure is used here.

We solve a least-squares inverse problem within each spherical cap. In the spherical cap's reference frame where $\alpha = 6,371.2$ km is the Earth's reference radius, *r* is the radius, θ is the colatitude, and ϕ is the longitude, the R-SCHA mathematical basis functions express the magnetic potential *V* as a sum of the Legendre (V₁) and Mehler (V₂) potentials

$$V(\boldsymbol{r},\boldsymbol{\theta},\boldsymbol{\varphi}) = V_1(\boldsymbol{r},\boldsymbol{\theta},\boldsymbol{\varphi}) + V_2(\boldsymbol{r},\boldsymbol{\theta},\boldsymbol{\varphi})$$

where the Legendre potential writes



$$V_1(r,\theta,\phi) = \alpha \sum_{k\geq 1-k\leq m\leq k} \left[\left(\frac{\alpha}{r}\right)^{n_k+1} G_{n_k}^{i,m} + \left(\frac{r}{\alpha}\right)^{n_k} G_{n_k}^{e,m} \right] \beta_{n_k}^m(\theta,\phi),$$

with $\beta_{n_k}^m(\theta,\phi)$ the associated Legendre harmonic functions with integer orders *m* and real degree n_k , and $G_{n_k}^{i,m}$ and $G_{n_k}^{e,m}$ the regional Legendre parameters to be estimated.

The Mehler potential is defined by

$$V_2(\mathbf{r},\boldsymbol{\theta},\boldsymbol{\phi}) = \boldsymbol{\alpha} \sum_{p \ge 0} \sum_{m \ge 0} \boldsymbol{G}_p^m \boldsymbol{\gamma}_p^m(\mathbf{r},\boldsymbol{\theta},\boldsymbol{\phi}),$$

where $\gamma_p^m(r, \theta, \phi)$ are the Mehler basis functions and G_p^m the unknown parameters.

Note that when considering such regional modeling, both sets of parameters G_p^m and $G_{n_k}^{e,m}$ are needed to account for magnetic field sources creating a field not fulfilling the null condition at the edges of the modeled region. This generally happens because magnetic field sources outside the modeled region contribute to the magnetic field measured inside this region (see Thébault et al., 2006).

The magnetic field (by convention the negative gradient of potential *V*) is solved within caps of 8° half aperture (about 1,800 km diameter at Earth's mean radius) and between the minimum and maximum available data altitudes in each region. The maximum expansion index *k* in the potential V_1 depends on the data availability on the cap's horizontal surface and the *p* index in the potential V_2 depends on the data availability with the altitude. We therefore restrict the Mehler's solution to a maximum index p = 4 and the Legendre's solution to a maximum index k = 150, for consistency with the near-surface and satellite data coverage on the sphere and with their altitude. We also select only degrees $n_k \leq 1100$ for the $G_{n_k}^{e,m}$ coefficients since the latter parameters mostly carry the wavelengths larger than the aperture of the cap (larger than about 1,800 km). The maximum degree n_k being related to the minimum horizontal spatial wavelength ($\lambda \sim 2\pi\alpha / n_k$), the maximum horizontal resolution of the model is then about 40 km at Earth's mean radius.

For the construction of the R-SCHA basis functions, we apply a procedure designed to both optimize the overall computation time and minimize the *a priori* large-scale satellite-based information contained in the WDMAM-2 anomaly field data. We rely on the statistical equivalence between the real-valued R-SCHA degrees n_k and the integer SH degree *n*. Since the CHAMP and Swarm satellite measurements are available between 250 and 460 km altitude, the signal-to-noise ratio at length scales smaller than these altitudes is low. To account for this, R-SCHA basis functions corresponding to $n_k > 300$ are not computed at satellite altitudes (typically corresponding to spatial structures smaller than about 125 km; that is, half of the minimum altitude found to be 250 km in the CHAMP satellite measurements). This approximation saves a significant amount of computational time. It was fully tested on synthetic simulations and is also justified considering the decrease of the Legendre potential values with increasing radius *r*.

The analogy between the standard SH and the R-SCHA degrees is further used to downweight the largescale GRIMM_L120 lithospheric field model contribution present by construction in the WDMAM-2 scalar anomaly data. This is done to avoid unnecessarily constraining our model by the information provided by GRIMM_L120. We define a weight function for the Legendre basis functions with a cutoff wavelength $\lambda_c = 800$ km (i.e., SH degree 100) using a high-pass Butterworth filter of order k = 6

$$w(\lambda_k) = \frac{1}{\sqrt{1 + (\lambda_k / \lambda_c)^k}}.$$

At R-SCHA degrees $n_k \sim 15$ (i.e., $\lambda_k \sim 2670$ km) this weight is equal to 0.5 and at degrees $n_k \sim 100$ (i.e., $\lambda_k \sim 400$ km) it reaches 1. We multiply the Legendre basis functions computed at the near-surface scalar altitudes by the weight values $w(\lambda_k)$ for each degree n_k . In contrast, the basis functions computed at satellite altitudes are not downweighted, to allow CHAMP and Swarm data to constrain the large lithospheric field wavelengths. The WDMAM-2 scalar anomaly grid is therefore downweighted in the GRIMM_L120 spectral band and smoothly weighted toward unity at the equivalent SH degree 100, beyond which the grid no longer

contains the *a priori* GRIMM_L120 truncated model. Completely canceling out the WDMAM-2 contributions in the GRIMM_L120 spectral band (i.e., setting $w(\lambda_k) = 0$ down to $\lambda_k \sim 400$ km) was not possible as this was found to lead to instabilities in the inverse problem.

For the data inversion, scalar gradients and scalar anomaly data are linearized along the Earth's main field. We follow Lesur et al. (2016) and linearize the WDMAM-2 scalar anomaly data at each location along the CM4 vector main field model (Sabaka et al., 2004) for the 1990 reference epoch of WDMAM-2. The CHAMP and the Swarm magnetic field scalar and scalar gradient data are linearized along the CHAOS-7.5 time-varying core field model (Finlay et al., 2020) up to SH degree 15 at each data location and corresponding epoch between 2000 and 2020.

A regional regularization is applied in the inverse problem. Indeed, Backus (1970) showed that uniqueness is not guaranteed when trying to recover the internal geomagnetic vector field from just the knowledge of its intensity. When producing models, this leads to a "Backus effect" responsible for noise amplification in the direction perpendicular to the main field, particularly near the magnetic equator (e.g., Khokhlov et al., 1997). Global models involving SH representation usually rely on a global regularization (e.g., see Equation 39 in Maus, 2010). Here, however, the regional approach used allows us to account for the Backus effect at a regional scale and to minimize the crustal field model components that are perpendicular to the main ambient field only for spherical caps whose centers are located within a magnetic latitude band $\pm 40^{\circ}$ around the magnetic equator.

For the numerical least-squares inversion, we apply a reweighting scheme using Huber weights to minimize the leverage of discrepant magnetic field measurements (Constable, 1988). However, the usual reweighting scheme involving many iterations was not found to be optimal when merging the different data types (and data altitudes). The reason for this is that the WDMAM-2 data contains information at horizontal wavelengths smaller than 40 km that does not comply with the Huber statistics (based on a mixture of Gaussian and Laplacian error distributions). In the full iterative reweighting scheme, this leads to regional models with artificially low power in regions where the near-surface magnetic anomalies have complex and strong signal at length scales smaller than 40 km. In practice, a maximum number of two iterations was found to be more appropriate for preserving the signal amplitude and resolution at Earth's mean radius.

Each of the 700 R-SCHA models is described by 5,379 regional parameters. They can be derived independently, allowing parallelization, independent assessment, and testing of each regional solution. This also saves computational and memory resources otherwise hardly manageable in a single global SH inversion. The series of regional R-SCHA models expressed at Earth's mean radius was finally transformed into a consistent set of SH Gauss coefficients to degree and order 1100 (using Thébault et al., 2016). However, the model is released only up to SH degree 1050, considering that the last harmonics may be contaminated by spectral leakage of structures smaller than the maximum model resolution.

4. Results and Discussion

The satellite data are successfully fitted by the regional models to a root mean square residuals lower than 4 nT in polar areas (Table 1). These residuals reflect the presence of remaining transient external perturbation field in the preprocessed data, despite the carefully applied dedicated correction. A few large-scale residual structures also remain at mid-latitudes, following the magnetic equator, as a result of the more conventional and less stringent correction in terms of low SH functions. The mean residuals for the CHAMP intensity data are slightly biased toward negative values while Swarm intensity data residuals are close to zero. This peculiarity has already been observed when combining data from different satellite missions (see, e.g., Finlay et al., 2020, their Tables 6 and 7) and could result from the way their instruments are calibrated in space.

Figures 1a and 1b show example scalar anomaly fields predicted by our final global model over Australia and the North Atlantic Ocean, as well as residuals between these model predictions and the original WD-MAM-2 grid. Residuals are mostly located over complex crustal magnetic field structures containing spatial wavelengths smaller than 40 km. These near-surface residuals also display large-scale structures testifying for disagreements between our model and the GRIMM_L120 lithospheric field model used to complement





Figure 1. (a) Scalar anomaly predictions of the present model (left) and residual map with respect to the WDMAM-2 grid (right) for Australia at Earth's mean radius; projection is Mercator. (b) Same as in panel (a) but over the North Atlantic region; the shadowed gray patches (in the right panel) indicate where the magnetization map of Dyment et al. (2015) was used to fill the WDMAM-2 data gap (after the data index map of Lesur et al., 2016). (c) Radial component of the present SH model at Earth's mean radius up to degree 1050; projection is Mollweide. All figures are plotted using the same color scale in nT units.

ground data in the WDMAM-2 grid. Some of these features are clearly apparent over Australia but also offshore of North Africa and Spain, in the form of North-South stripes with an East-West change of polarity. Such structures are known to be artifacts related to external field contamination when using satellite measurements (Thébault et al., 2012, 2017). The fact that they show up in our residual maps confirms that these GRIMM_L120 artifactual structures do not comply with the CHAMP and additional Swarm satellite measurements we used and that they are properly rejected by our model. This a posteriori validates our procedure to downweight contributions from GRIMM_L120 when using the WDMAM-2 grid. Over oceans, residuals are locally larger and structured along oceanic chrons. Figure 1b illustrates this point. It displays magnetic field structures and residuals with sharp transitions over the North Atlantic region. Some of these

transitions are well covered and documented by marine and airborne measurements but not captured by our model. It is for instance the case over the Reykjanes ridge, south of Iceland, and down to the Bight oceanic transform fault. Accounting for these would require a higher resolution SH model. In other regions, such as those highlighted using light gray shading in Figure 1b, residuals also reveal small-scale patterns. They correspond to additional information added to the WDMAM-2 grid to fill oceanic data gaps using the magnetization model of Dyment et al. (2015). Significant residuals in oceans are globally well correlated with the location of these synthetic data because such discontinuous and geometrical patterns are difficult to resolve with continuous mathematical functions.

A map of the radial component predicted at Earth's mean radius by our SH vector lithospheric field is finally shown in Figure 1c. Magnitude is generally weaker near the equator than near the pole, as expected and as had been observed in the WDMAM-2 scalar anomaly grid. This is because the lithospheric field induced by the Earth's main field dominates in the continental crust and increases with absolute latitude (Maus & Haak, 2002).

Figure 2a shows the power spectrum of the SH model plotted at Earth's surface between SH degrees 16 and 185. This range allows visual comparisons with the spectra of the MF7 model derived from the final 2 years of the CHAMP data (Maus et al., 2008), the LCS-1 model derived from the CHAMP and Swarm satellite observations (Olsen et al., 2017), and the WDMAM-2 model (Lesur et al., 2016). Between SH degrees 16 and 100, the present model closely follows the MF7 and LCS-1 model and deviates from the WDMAM-2 model. From SH degrees 100 onwards, the model then closely follows the WDMAM-2 model. This behavior is what was expected and further suggests that the model properly accounts for both the satellite and near-surface data we used.

Figure 2b displays the power spectrum of our model over its entire range, which we now compare to the spectra of the NGDC-720 model derived from the EMAG2 grid (Maus, 2010), of the Enhanced Magnetic Model (EMM-2017) computed from the EMAG2v3 grid (Meyer et al., 2017) and of the WDMAM-2 model (Lesur et al., 2016). All spectra show some varying amount of power dip between SH degree 120 and about 200. This results from the lack of constraints provided by the satellite and near-surface data in this degree range. For larger degrees, all spectra have comparable levels up to SH degree 300, beyond which their slopes significantly differ. Though derived from different datasets and ocean models, the NGDC-720 and the WD-MAM-2 display similar spectra up to about SH degree 600, before becoming flat. In contrast, the spectrum of our model exhibits a quasi-constant negative slope up to SH degree 830 beyond which the slope becomes steeper. A decreasing, rather than flat, power spectrum is consistent with remaining data gaps and testifies for the stability of the model.

We also tested our model against a statistical representation of its spectrum between degrees 16 and 1050 (shown in black with its 95% Confidence Interval [CI] in Figure 2b). The statistical model is parameterized in terms of a mean apparent crustal magnetization, a mean magnetic crustal thickness, and a power-law related to the magnetic sources, following Thébault and Vervelidou (2015). The least-squares fit between the observed and the statistical spectra leads to a mean magnetization estimate of 0.4 ± 0.3 A m⁻¹ (95% CI), a mean magnetic thickness equal to 36 ± 6.5 km (95% CI) and a power-law equal to -1.36 ± 0.32 (95% CI). These estimates are realistic and in agreement with independent geophysical values (see Thébault & Vervelidou, 2015). The observed power spectrum lies outside the 95% CI of the statistical fit only for SH degrees beyond 830, where the observed power spectrum has a steeper slope because of data gaps.

The model presented in this work is the first successful lithospheric magnetic field model derived up to SH degree 1050 and based on a joint inversion of publicly available near-surface global compilations and CHAMP and Swarm satellite data. Using the same approach, additional models with increasing resolution could next be built as more satellite and near-surface measurements become available. An important target will be to constrain the spectral gap between SH degrees 120 and 180. Different scenarios are currently explored for the evolution of the ongoing Swarm mission. Measuring the magnetic field scalar and vector gradients at steadily decreasing altitudes during the Swarm A and C satellites re-entry will provide invaluable low altitude measurements for filling this gap. In parallel, additional efforts to compile and add more marine and airborne measurements in global compilations such as the WDMAM are also much needed to



Geophysical Research Letters



Figure 2. Spherical Harmonic power spectra. (a) Low degree (up to SH degree 185) spectra of the new SH model (red), of the MF7 model (black, Maus et al., 2008), of the LCS-1 model (blue, Olsen et al., 2017), and of the WDMAM-2 model (green, Lesur et al., 2016); (b) Full spectra of the new SH model (up to SH degree 1050, red), of the NGDC-720 model (up to SH degree 720, blue, Maus, 2010), of the WDMAM-2 model (up to SH degree 800, green, Lesur et al., 2016) and of the EMAG2v3/EMM-2017 model (up to SH degree 790, pink, Meyer et al., 2017). The black curves are the maximum likelihood of a statistical model and its 95% CI best fitting the new SH model using the statistical expression of Thébault and Vervelidou (2015).

improve the robustness of the still poorly constrained intermediate lithospheric field wavelengths in these compilations.

Data Availability Statement

The Swarm data are available at ftp://swarm-diss.eo.esa.int/Level1b/Entire_mission_data/MAGx_LR/ or by any web browser at https://swarm-diss.eo.esa.int/#swarm%2FLevel1b%2FEntire_mission_data%2FMAGx_LR. The World Digital Magnetic Anomaly Map data are available at www.wdmam.org. The CHAOS-7.5 core

field model is available at https://www.spacecenter.dk/files/magnetic-models/CHAOS-7/. The lithospheric field model to degree 1050 presented in this study is available in spherical harmonics at the Zenodo CERN's Data Center https://zenodo.org/record/5546528.

References

Backus, G. E. (1970). Non-uniqueness of the external geomagnetic field determined by surface intensity measurements. *Journal of Geophysical Research*, 75(31), 6339–6341. https://doi.org/10.1029/ja075i031p06339

- Chulliat, A., Vigneron, P., Thébault, E., Sirol, O., & Hulot, G. (2013). Swarm SCARF dedicated ionospheric field inversion chain. *Earth Planets and Space*, 65(11), 1271–1283. https://doi.org/10.5047/eps.2013.08.006
- Constable, C. G. (1988). Parameter estimation in non-Gaussian noise. *Geophysical Journal*, 94, 131–142. https://doi.org/10.1111/j.1365-246X.1988.tb03433.x
 - Dyment, J., Choi, Y., Hamoudi, M., Lesur, V., & Thébault, E. (2015). Global equivalent magnetization of the oceanic lithosphere. Earth and Planetary Science Letters, 430, 54–65. https://doi.org/10.1016/j.epsl.2015.08.002
 - Finlay, C. C., Kloss, C., Olsen, N., Hammer, M. D., Tøffner-Clausen, L., Grayver, A., & Kuvshinov, A. (2020). The CHAOS-7 geomagnetic field model and observed changes in the South Atlantic Anomaly. *Earth Planets and Space*, 72, 156. https://doi.org/10.1186/ s40623-020-01252-9
 - Friis-Christensen, E., Lühr, H., & Hulot, G. (2006). Swarm: A constellation to study the Earth's magnetic field. *Earth Planets and Space*, 58(4), 351–358. https://doi.org/10.1186/bf03351933

GFZ Section 2.3. (2019). CHAMP: Overview of final ME products and format description (Scientific Technical Report STR—Data; 19/10). GFZ German Research Centre for Geosciences. https://doi.org/10.2312/GFZ.b103-19104

- Khokhlov, A., Hulot, G., & Le Mouël, J. L. (1997). On the Backus effect—I. *Geophysical Journal International*, 130(3), 701–703. https://doi.org/10.1111/j.1365-246X.1997.tb01864.x
- Korhonen, J., Fairhead, J. D., Hamoudi, M., Hemant, K., Lesur, V., Mandea, M., et al. (2007). Magnetic anomaly map of the world—Carte des anomalies magnétiques du monde (1st ed.). Commission for Geological Map of the World.
- Kovalevsky, J., Mueller, I. I., & Kolaczek, B. (Eds.), (2012). Reference frames: In astronomy and geophysics (Vol. 154). Springer Science & Business Media.
- Lesur, V., Hamoudi, M., Choi, Y., Dyment, J., & Thébault, E. (2016). Building the second version of the world digital magnetic anomaly map (WDMAM). Earth Planets and Space, 68(1), 27. https://doi.org/10.1186/s40623-016-0404-6
- Lesur, V., Rother, M., Vervelidou, F., Hamoudi, M., & Thébault, E. (2013). Post-processing scheme for modeling the lithospheric magnetic field. Solid Earth, 4, 105–118. https://doi.org/10.5194/sed-4-105-2013
- Maus, S. (2010). An ellipsoidal harmonic representation of Earth's lithospheric magnetic field to degree and order 720. Geochemistry, Geophysics, Geosystems, 11, Q06015. https://doi.org/10.1029/2010GC003026
- Maus, S., Barckhausen, U., Berkenbosch, H., Bournas, N., Brozena, J., Childers, V., et al. (2009). EMAG2: A 2-arc min resolution Earth Magnetic Anomaly Grid compiled from satellite, airborne, and marine magnetic measurements. *Geochemistry, Geophysics, Geosystems*, 10, Q08005. https://doi.org/10.1029/2009GC002471
- Maus, S., & Haak, V. (2002). Is the long wavelength crustal magnetic field dominated by remanent magnetisation? Journal of the Indian Geophysical Union, 6, 1–5.
- Maus, S., Sazonova, T., Hemant, K., Fairhead, J. D., & Ravat, D. (2007). National geophysical data center candidate for the world digital magnetic anomaly map. *Geochemistry, Geophysics, Geosystems*, 8(6). https://doi.org/10.1029/2007gc001643
- Maus, S., Yin, F., Lühr, H., Manoj, C., Rother, M., Rauberg, J., et al. (2008). Resolution of direction of oceanic magnetic lineations by the sixth generation lithospheric magnetic field model from CHAMP satellite magnetic measurements. *Geochemistry, Geophysics, Geosystems*, 9(7). https://doi.org/10.1029/2008gc001949
- Meyer, B., Chulliat, A., & Saltus, R. (2017). Derivation and error analysis of the earth magnetic anomaly grid at 2 arc min resolution version 3 (EMAG2v3). *Geochemistry, Geophysics, Geosystems, 18*(12), 4522–4537. https://doi.org/10.1002/2017gc007280
- Olsen, N., Ravat, D., Finlay, C. C., & Kother, L. K. (2017). LCS-1: A high-resolution global model of the lithospheric magnetic field derived from CHAMP and Swarm satellite observations. *Geophysical Journal International*, 211(3), 1461–1477. https://doi.org/10.1093/ gji/ggx381
- Purucker, M., Langlais, B., Olsen, N., Hulot, G., & Mandea, M. (2002). The southern edge of cratonic North America: Evidence from new satellite magnetometer observations. *Geophysical Research Letters*, 29(15), 56. https://doi.org/10.1029/2001GL013645
- Purucker, M. E., & Clark, D. A. (2011). Mapping and interpretation of the lithospheric magnetic field. In Geomagnetic observations and models (pp. 311–337). Springer. https://doi.org/10.1007/978-90-481-9858-0_13
- Ravat, D., Whaler, K. A., Pilkington, M., Sabaka, T., & Purucker, M. (2002). Compatibility of high-altitude aeromagnetic and satellite-altitude magnetic anomalies over Canada. *Geophysics*, 67(2), 546–554. https://doi.org/10.1190/1.1468615
- Reigber, C., Lühr, H., & Schwintzer, P. (2002). CHAMP mission status. Advances in Space Research, 30(2), 129–134. https://doi.org/10.1016/s0273-1177(02)00276-4
- Richardson, I. G., & Cane, H. V. (2011). Geoeffectiveness (Dst and Kp) of interplanetary coronal mass ejections during 1995–2009 and implications for storm forecasting. *Space Weather*, 9(7). https://doi.org/10.1029/2011sw000670
- Richmond, A. D. (1995). Ionospheric electrodynamics using magnetic Apex coordinates. *Journal of Geomagnetism and Geoelectricity*, 47, 191–212. https://doi.org/10.5636/jgg.47.191
- Sabaka, T. J., Olsen, N., & Purucker, M. E. (2004). Extending comprehensive models of the Earth's magnetic field with Ørsted and CHAMP data. *Geophysical Journal International*, 159(2), 521–547. https://doi.org/10.1111/j.1365-246x.2004.02421.x
- Thébault, E. (2006). Global lithospheric magnetic field modelling by successive regional analysis. *Earth Planets and Space*, 58(4), 485–495. https://doi.org/10.1186/bf03351944
- Thébault, E., Lesur, V., Kauristie, K., & Shore, R. (2017). Magnetic field data correction in space for modelling the lithospheric magnetic field. *Space Science Reviews*, 206(1–4), 191–223. https://doi.org/10.1007/s11214-016-0309-5
- Thébault, E., Schott, J. J., & Mandea, M. (2006). Revised spherical cap harmonic analysis (R-SCHA): Validation and properties. *Journal of Geophysical Research: Solid Earth*, 111(B1). https://doi.org/10.1029/2005jb003836
- Thébault, E., & Vervelidou, F. (2015). A statistical spatial power spectrum of the Earth's lithospheric magnetic field. *Geophysical Journal International*, 201(2), 605–620. https://doi.org/10.1093/gji/ggu463

Acknowledgments

The European Spatial Agency (ESA) is gratefully acknowledged for providing access to the L1b Swarm data. The authors would like to thank J. Dyment and V. Lesur for making the World Digital Magnetic Anomaly map grid available online. This study was partly funded by CNES in the framework of the project "Exploitation de la mission spatiale Swarm" and by the ESA DISC through the project "Extending the dedicated lithospheric field inversion." The authors thank D. Ravat and S. Maus for helping us improving the manuscript and the editor M. Korte.



- Thébault, E., Vervelidou, F., Lesur, V., & Hamoudi, M. (2012). The satellite along-track analysis in planetary magnetism. *Geophysical Journal International*, 188(3), 891–907. https://doi.org/10.1111/j.1365-246x.2011.05281.x
- Thébault, E., Vigneron, P., Langlais, B., & Hulot, G. (2016). A Swarm lithospheric magnetic field model to SH degree 80. Earth Planets and Space, 68(1), 1–13. https://doi.org/10.1186/s40623-016-0510-5
- Thébault, E., Vigneron, P., Maus, S., Chulliat, A., Sirol, O., & Hulot, G. (2013). Swarm SCARF dedicated lithospheric field inversion chain. *Earth Planets and Space*, 65(11), 1257–1270. https://doi.org/10.5047/eps.2013.07.008